Author's response to reviews

Title: The Association of Alcohol Consumption with Mammographic Density in a Multiethnic Urban Population

Authors:

Zoe Quandt (zoequandt@gmail.com)
Julie D Flom (jfl2259@caa.columbia.edu)
Parisa Tehranifar (pt140@cumc.columbia.edu)
Diane Reynolds (Diane.Reynolds@liu.edu)
Mary Beth Terry (mt146@columbia.edu)
Jasmine A McDonald (jam2319@columbia.edu)

Version: 2 Date: 29 August 2014

Author's response to reviews:

The article has not undergone peer review; therefore, extensive revisions and a response to reviewers are not provided in this submission.